

Clinical and prognostic significance of central and obstructive apnoeas in patients with transthyretin cardiac amyloidosis

Francesco Gentile ()^{1,2}, Alberto Giannoni^{1,2}, Alberto Aimo^{1,2}, Vincenzo Castiglione ()^{1,2}, Francesca Bramanti², Giovanni Iudice², Eleonora Degl'Innocenti², Michele Emdin ()^{1,2}*, Giuseppe Vergaro^{1,2}, and Claudio Passino ()^{1,2}

¹Health Science Interdisciplinary Center, Scuola Superiore Sant'Anna, Piazza Martiri della Libertà, 56127 Pisa, Italy; and ²Cardio-thoracic Department, Fondazione Toscana Gabriele Monasterio, via G. Moruzzi 1, 56124 Pisa, Italy

Received 2 July 2024; revised 6 August 2024; accepted 12 September 2024; online publish-ahead-of-print 23 September 2024

Aims	Central apnoea (CA) and obstructive apnoea (OA) are highly prevalent in patients with chronic heart failure (HF), and trans- thyretin cardiac amyloidosis (ATTR-CA) is an increasingly recognized HF aetiology. This study aims to investigate the preva- lence and impact of CA and OA in patients with ATTR-CA.
Methods and results	Consecutive patients with ATTR-CA who underwent 24 h ambulatory cardiorespiratory monitoring were enlisted for an evaluation of the prevalence and severity of breathing disorders. The severity of these disorders was quantified using the apnoea–hypopnoea index (AHI). Accordingly, the patients were categorized as having normal breathing (NB, AHI <5 events/h), OA (AHI \geq 5 events/h with >50% being obstructive), or CA (AHI >5 events/h with \geq 50% being central). The primary endpoint at follow-up was all-cause mortality. Out of 142 patients enrolled ($n = 142$, aged 77 ± 7 years, 91% males, 96% wild-type ATTR-CA), considering the 24 h monitoring, 20% had NB (39% at daytime and 8% at nighttime), 35% had CA (45% at daytime and 39% at nighttime), and 45% had OA (25% at daytime and 54% at nighttime). After a median 2.3-year (1.4–3.3 years) follow-up, 24 h, daytime, and nighttime AHIs were higher in non-survivors vs. survivors (all $P < 0.05$), independently of the prevalent apnoea type ($P = 0.64$). At multivariable regression analysis (adjusted for the possible clinical, echocardiographic, and biohumoral confounders), nighttime AHI \geq 30 events/h {hazard ratio 2.37 [95% confidence interval (CI) 1.07–5.23], $P = 0.033$ } and high-sensitivity troponin T [hazard ratio 2.43 (95% CI 1.42–4.17), $P = 0.001$] were predictors of mortality.
Conclusion	Both CA and OA are highly prevalent, both at daytime and nighttime, in patients with ATTR-CA and are associated with higher mortality.
Lay summary	This study investigated the prevalence and prognostic significance of central apnoea (CA) and obstructive apnoea (OA) in 142 patients with transthyretin cardiac amyloidosis. Both CA and OA were highly prevalent during the whole 24 h period, with only 20% were classified as having normal breathing [meant as an apnoea–hypopnoea index (AHI) <5 events/h during the 24 h period]. Obstructive apnoea was more frequent than CA, particularly during the night, while the prevalence of CA increased with worsening left ventricular systolic and diastolic dysfunctions. At follow-up, 24 h, daytime, and nighttime AHIs were higher in non-survivors vs. survivors, independently of the prevalent apnoea type, and at multivariable regression analysis (adjusted for the possible clinical, echocardiographic, and biohumoral confounders). nighttime AHI >30 events/h was an independent predictor of mortality.

* Corresponding author. Tel: +39 050 3152189, Email: m.emdin@santannapisa.it, emdin@ftgm.it

[©] The Author(s) 2024. Published by Oxford University Press on behalf of the European Society of Cardiology. All rights reserved. For commercial re-use, please contact reprints@oup.com for reprints and translation rights for reprints. All other permissions can be obtained through our RightsLink service via the Permissions link on the article page on our site—for further information please contact journals.permissions@oup.com.





Clinical and prognostic significance of central and obstructive apnoeas in patients with transthyretin cardiac amyloidosis. Central and obstructive apnoeas are highly prevalent in patients with transthyretin cardiac amyloidosis, and a severe apnoea/hypopnoea burden is associated with increased mortality. ATTR-CA, transthyretin cardiac amyloidosis; HR, hazard ratio; LVEF, left ventricular ejection fraction; NAC, National Amyloidosis Centre; NYHA, New York Heart Association; TAPSE, tricuspid annular plane systolic excursion.

Keywords Cardiac amyloidosis • Central apnoeas • Obstructive apnoeas • Periodic breathing • Cheyne–Stokes respiration

Introduction

Cardiac amyloidosis is a group of infiltrative cardiomyopathies secondary to the extracellular deposition of misfolded proteins.¹ Transthyretin cardiac amyloidosis (ATTR-CA) and immunoglobulin light-chain cardiac amyloidosis (AL-CA) are the two most common forms of cardiac amyloidosis.^{1,2}

Cardiac amyloidosis has been classified among restrictive cardiomyopathies, as it is characterized by wall stiffness and diastolic dysfunction secondary to amyloid infiltration and myocardial pseudohypertrophy.³ However, left ventricular (LV) systolic dysfunction often coexists.⁴ Accordingly, while cardiac amyloidosis is an increasingly recognized cause of heart failure (HF) with preserved LV ejection fraction (LVEF) (HFpEF, \geq 50%),⁵ it may also present as or evolve into HF with mildly reduced LVEF (HFmrEF, 41–49%) or reduced LVEF (HFrEF, \leq 40%).⁶

Following recent advances in diagnostic and therapeutic tools,⁷ ATTR-CA is increasingly diagnosed in the cardiovascular setting and is no longer considered a rare disease, as recently confirmed in a screening study, showing a 0.46% prevalence of ATTR-CA in subjects \geq 65 years.⁸ A deeper understanding of the pathophysiological mechanisms and prognostic drivers of the disease is hence warranted to improve

risk stratification and offer tailored therapies for patients with ATTR-CA.

Breathing disorders are highly prevalent in patients with HF and are associated with higher morbidity and mortality irrespective of LVEF.⁹ While obstructive apnoeas (OAs) are sleep-related phenomena, secondary to the collapse of the upper airways,¹⁰ central apnoeas (CAs) occur also in awake¹¹ and upright patients with HF,¹² elicited by increased chemoreflex sensitivity, plant gain, and circulatory delay.¹³ Furthermore, OAs are more frequently observed in HFpEF, while the prevalence and severity of CA increase with worsening LV systolic and diastolic dysfunctions.¹⁴

Because the clinical significance of breathing disorders has been scarcely investigated in patients with ATTR-CA,¹⁵ this study aimed to assess the prevalence, correlates, and prognostic impact of either day-time or nighttime breathing disorders in a contemporary cohort of patients with ATTR-CA.

Methods

Subjects and study design

Consecutive patients with a diagnosis of ATTR-CA referred to the outpatient clinic of the Fondazione Monasterio (Pisa, Italy) between April 2016 and June 2023 were prospectively enrolled to undergo cardiorespiratory monitoring to investigate the prevalence and severity of breathing disorders (as detailed in the following). Severe physical and/or cognitive impairment, severe pulmonary disease, and treatment with therapies potentially influencing breathing control [e.g. opioids, theophylline, oxygen, and continuous positive airway pressure (CPAP)] were considered as exclusion criteria.

As mentioned previously, 24 h ambulatory cardiorespiratory monitoring (Somté PSG2; Compumedics, Abbotsford, VIC, Australia) was used in the study and included nasal airflow, chest and abdominal respiratory movements by inductance plethysmography belts, and oxygen saturation by a pulse oximeter.¹⁶ As recommended by guidelines,¹⁷ apnoeas were defined as the cessation of airflow for ≥ 10 s and classified as either obstructive or central based on the presence/absence of respiratory movements. Hypophoeas were defined as a reduction in airflow \geq 50%, lasting for \geq 10 s, with a reduction in oxygen saturation \geq 4%. The severity of breathing disorders was quantified through the apnoea-hypopnoea index (AHI) for the whole 24 h period, with daytime defined as the period between 7:00 a.m. and 9:59 p.m. and nighttime as the period between 10:00 p.m. and 6:59 a.m..^{16,17} Apnoeic events during the daytime were considered only in the setting of periodicity, excluding pauses due to speaking, drinking, or eating. Patients with a 24 h AHI <5 event/h were classified as having normal breathing (NB), while patients with a 24 h AHI \geq 5 event/h were classified as having either CA or OA according to the prevalent phenotype (>50% of events being obstructive or central, respectively). As recommended,¹⁷ a severe apnoea burden was considered as an AHI \geq 30 events/h. Finally, the burden of oxygen desaturation was calculated as the time spent with a saturation <90% (T-90, minutes).

Beyond the cardiorespiratory monitoring, at enrolment, all patients underwent a standard transthoracic echocardiography (IE33 or Epiq7; Philips Medical Systems, Palo Alto, CA, USA) to assess chamber dimension and systolic and diastolic functions, according to intersociety guidelines.¹⁸ For biohumoral analysis, haemoglobin, estimated glomerular filtration rate (eGFR, CKD-EPI formula), N-terminal pro-B-type natriuretic peptide (NT-proBNP; Roche Diagnostics assay), and high-sensitivity troponin T (hs-troponin T; Roche Diagnostics assay) were collected from all patients. According to the eGFR and NT-proBNP values, patients were classified into three severity stages [i.e. National Amyloidosis Center (NAC) stages], as proposed by Gillmore et al.¹⁵

All patients were followed up at the outpatient clinic of Fondazione Monasterio until 31 December 2023 for the endpoint of all-cause death. The outcome status was determined from the medical records or telephone interviews with patients, relatives, or general practitioners.

Before enrolment, an informed consent was obtained from all patients involved in this study, which was approved by the Institutional Review Board Committee of Fondazione Monasterio and conducted in accordance with the Declaration of Helsinki of the World Medical Association.

Statistical analysis

Quantitative values were presented as mean ± standard deviation or median [inter-quartile range (IQR)] according to distribution, while qualitative values as numbers and/or percentages.

Clinical, biohumoral, echocardiographic, and cardiorespiratory monitoring parameters were compared among patients classified as having NB, CA, or OA. The mean differences were compared by a one-way analysis of variance or Kruskal-Wallis test as appropriate, with a Bonferroni post hoc correction. Discrete variables were compared by the χ^2 test with Yates's correction. A Pearson's correlation analysis was performed to identify the clinical characteristics associated with a severity of breathing disorders.

For survival analysis, patients were dichotomized according to the presence of severe breathing disorders (i.e. AHI ≥30 events/h) during the whole 24 h period, at daytime, or at nighttime. Accordingly, the Kaplan-Meier method with log-rank statistics was used to estimate survival. The prognostic significance of CA vs. OA was compared as well.

The independent prognostic values of 24 h, daytime, and nighttime AHI were determined by a stepwise multivariable Cox proportional hazards model, adjusted for other well-known risk predictors in patients with ATTR-CA, namely patient age, atrial fibrillation, New York Heart Association (NYHA) class, NAC stage, hs-troponin T, treatment with tafamidis or continuous positive airway pressure (CPAP), E/e' average, LVEF, and tricuspid annular plane systolic excursion (TAPSE). Significant univariable predictors (P < 0.05) were considered as possible confounders and included in the multivariable analysis, while non-significant variables were omitted in a stepwise fashion, according to the P-values. Non-normally distributed variables were log-transformed before entering regressions.

A statistical analysis was performed by using R statistical software (version 3.4.0, The R Foundation for Statistical Computing, Vienna, Austria). A two-tailed P-value < 0.05 was considered significant.

Results

Study population

The study population consisted of 142 patients (mean age 77 \pm 7 years, 91% males, 96% with wild-type ATTR-CA, and 4% with hereditary ATTR-CA). Most patients complained of dysphoea, 23, 53, and 24% being classified as NYHA Classes I, II, and III, respectively. The mean LVEF was 51% (9%), with 62, 19, and 19% classified as HFpEF, HFmrEF, and HFrEF, respectively. The mean eGFR was 62 (18) mL/min/1.73 m², and the median NT-proBNP was 2203 (IQR 850-4853) ng/L. Accordingly, 78 (55%), 45 (32%), and 19 (13%) patients were classified as having NAC Stages I, II, and III, respectively. Most patients had diastolic dysfunction, classified as Grades II and III in 37 and 34% patients, respectively, while 36 (25%) patients had atrial fibrillation at the time of examination. At enrolment, 19 (13%) patients were treated with tafamidis for a median of 7-month period (range 1–32 months).

Prevalence, severity, and clinical correlates of breathing disorders

During the 24 h period, breathing disorders were highly prevalent in the study population, with only 20% of patients (39% at daytime and 8% at nighttime) identified as having NB (Figure 1). While OA was the most common phenotype during the 24 h period (45%), the prevalence of CA (35%) increased with worsening LV systolic and diastolic functions (Figure 1). Furthermore, while OAs were more prevalent and severe at nighttime (53%, with 29% of patients having an AHI \geq 15 events/h), the prevalence of CA was similar at daytime (36%, with 17% of patients having an AHI \geq 15 events/h) and at nighttime (39%, with 27% of patients having an AHI \geq 15 events/h; *Figure* 2). No significant differences were observed among groups as for clinical features and therapies (Table 1), though patients with CA had more dilated left atria and LV and lower LVEF than patients with OA (Table 2). Compared with patients with OA, those with CA had a higher AHI at both daytime and nighttime and a higher T-90 (Table 2). Detailed data about the number of apnoea/hypopnoea events at daytime, at nighttime, and during the 24 h period and about oxygen saturation are reported in Supplementary material online, Table S1. As reported in Supplementary material online, Table S2, no significant differences were observed when comparing the prevalence and severity of breathing disorders among patients with a different Perugini scoring system at nuclear imaging.

Daytime AHI was associated with worse symptoms, higher biomarker levels, worsening LV systolic and diastolic functions, and mitral regurgitation severity (Table 3), whereas nighttime AHI was significantly



Figure 1 Prevalence of breathing disorders in patients with transthyretin cardiac amyloidosis according to the severity of either systolic or diastolic dysfunction. Patients with transthyretin cardiac amyloidosis show high prevalence of breathing disorders during the 24 h period. Notably, the prevalence of central apnoeas increases with worsening left ventricle systolic and diastolic dysfunctions.



Figure 2 Severity of breathing disorders at daytime and at nighttime in patients with transthyretin cardiac amyloidosis. In patients with transthyretin cardiac amyloidosis, both central apnoea and obstructive apnoea were observed during the 24 h period, although they were more prevalent and severe at nighttime.

associated with NT-proBNP and hs-troponin T levels and mitral regurgitation severity (*Table 3*).

Prognostic significance of breathing disorders

During a median 2.3-year (IQR 1.4–3.3 years) follow-up, 27 (19%) patients died. Compared with survivors, non-survivors had a significantly higher AHI at daytime, at nighttime, and during the 24 h period (*Figure 3*). As shown in the Kaplan–Meier curves in *Figure 4*, severe breathing disorders during the whole 24 h period, at daytime, or at nighttime were significantly associated with a higher mortality. No difference was observed when comparing patients with CA vs. OA according to the prevalent apnoea phenotype (log-rank 0.22, P = 0.64).

At adjusted regression analysis, only nighttime AHI \geq 30 events/h {hazard ratio 2.37 [95% confidence interval (Cl) 1.07–5.23], *P* = 0.033} and log-transformed hs-troponin T [hazard ratio 2.43 (95%Cl 1.42–4.17), *P* = 0.001] remained as independent predictors of mortality.

Discussion

The prevalence and clinical significance of daytime and nighttime breathing disorders in patients with ATTR-CA were investigated for

Variables	All patients (n = 142)	Normal breathing (n = 28) (20%)	Obstructive apnoeas (n = 64) (45%)	Central apnoeas (n = 50) (35%)
Clinical features				
Age, years	77 <u>+</u> 7	77 <u>+</u> 6	77 ± 6	77 <u>±</u> 8
Men, <i>n</i> (%)	129 (91)	23 (82)	59 (92)	47 (94)
BMI, kg/m ²	28 ± 4	27 ± 3	28 ± 4	28 ± 4
NYHA III, n (%)	32 (24)	7 (25)	14 (23)	11 (24)
Atrial fibrillation, n (%)	36 (28)	9 (38)	13 (22)	14 (32)
NT-proBNP, ng/L	2203 (850–4853)	1969 (719–3579)	2203 (1057–4550)	2262 (884–5780)
eGFR, mL/min/1.73 m ² (SD)	62 <u>+</u> 18	63 <u>+</u> 19	61 ± 18	62 ± 20
NAC Stage I, n (%)	78 (55)	17 (61)	37 (58)	24 (48)
NAC Stage II, n (%)	45 (32)	6 (21)	21 (33)	(18 36)
NAC Stage III, n (%)	13 (19)	5 (18)	6 (9)	8 (16)
Hs-troponin T, ng/L	44 (32–66)	36 (27–53)	48 (33–71)	45 (32–62)
Comorbidities				
Hypertension, n (%)	87 (61)	14 (50)	42 (66)	31 (62)
Diabetes, n (%)	25 (18)	6 (21)	10 (16)	9 (18)
CAD, n (%)	34 (24)	4 (14)	17 (27)	13 (26)
COPD, n (%)	15 (11)	1 (4)	9 (14)	5 (10)
Hb, g/dL (SD)	14 (2)	13 (2)	14 (2)	14 (2)
Baseline treatment				
Tafamidis, n (%)	19 (13)	3 (11)	12 (19)	4 (8)
Beta-blockers, n (%)	92 (79)	22 (85)	35 (59)	35 (75)
ACEi/ARB, n (%)	71 (54)	15 (58)	30 (51)	26 (55)
ARNI, n (%)	6 (5)	1 (4)	2 (3)	3 (6)
MRA, n (%)	38 (29)	9 (35)	17 (29)	12 (26)
SGLT2i, n (%)	13 (10)	4 (15)	5 (9)	4 (9)
Furosemide, n (%)	85 (64)	16 (62)	37 (63)	32 (68)
PM, n (%)	9 (7)	2 (7)	3 (5)	4 (8)
ICD, n (%)	7 (5)	4 (14)	1 (2)	2 (4)
CRT, n (%)	1 (1)	0 (0)	0 (0)	1 (2)

Table 1	Characteristics of	f the study	popula	ation accordin	g to the	breathing p	atterns

Values are mean \pm SD, median (inter-quartile range), or n (%).

ACEi, angiotensin converting-enzyme inhibitors; ARB, angiotensin receptor blockers; ARNI, angiotensin receptor–neprilysin inhibitor; BMI, body mass index; CAD, coronary artery disease; COPD, chronic obstructive pulmonary disease; CRT, cardiac resynchronization therapy; eGFR, estimated glomerular filtration rate; Hb, haemoglobin; ICD, implantable cardioverter–defibrillator; MRA, mineralocorticoid receptor antagonists; NAC, National Amyloidosis Center; NT-proBNP, N-terminal pro-B-type natriuretic peptide; NYHA, New York Heart Association; PM, pacemaker; SGLT2i, sodium–glucose cotransporter-2 inhibitors.

the first time. Both CA and OA were highly prevalent in this subset, and similar to other HF cohorts, a severe apnoea/hypopnoea burden was a predictor of mortality independent of possible confounders.

To the best of our knowledge, only a single study has been previously reported on sleep-disordered breathing in patients with cardiac amyloidosis, which included 39 patients with ATTR-CA (22 mutated and 17 wild-type) and 31 patients with AL-CA who underwent an overnight unattended polygraphy.¹⁵ Conversely, our study predominantly focused on patients with wild-type ATTR-CA undergoing 24 h cardiorespiratory monitoring. Indeed, considering the clinical and prognostic differences between patients with ATTR-CA and those with AL-CA, including both groups may have influenced survival analysis. Despite these differences, in both populations OAs were more prevalent than CAs and the prevalence of nighttime breathing disorders (90%) in the previous study¹⁵ was very close to the 92% observed in our larger cohort. In this respect, the epidemiological characteristics of breathing disorders from those reported in patients with ATTR-CA do not seem to be different from those reported in patients with HF across the LVEF spectrum.⁹ Indeed,

at least during nighttime, OAs were more frequent in patients with HFpEF, while the prevalence of CA increased with worsening LVEF in line with previous findings in a large population (n = 700) of patients with HF.¹⁴ This is not surprising considering that LV systolic dysfunction directly affects circulation time, which is a key pathophysiological determinant of CA.²⁰ Furthermore, in the present study, diastolic dysfunction and left atrium enlargement were associated with a higher prevalence of CA. Chemoreflex sensitization, secondary to increased filling pressure, may be, in this case, a potential mechanism, as it has been demonstrated in HF.²¹

A higher AHI was an independent predictor of outcomes in this population, consistent with previous studies in populations with HF.^{11,22–24} Conversely, oxygen desaturation, as expressed by the T-90, was not a predictor of mortality in the current study. This finding seems contradictory to what was observed in the study by Bodez *et al.*,¹⁵ in which nocturnal desaturation, but not AHI, was an independent predictor of outcomes in patients with either AL-CA or ATTR-CA. However, as noted by the same authors, including patients with AL-CA

Variables	All patients, n = 142	Normal breathing, <i>n</i> = 28 (20%)	Obstructive apnoeas, $n = 64$ (35%)	Central apnoeas, <i>n</i> = 50 (45%)
Echocardiography				
LAVi, mL/m ²	41 <u>+</u> 9	40 ± 9	39 ± 8	44 \pm 11 $^+$
Severe MR, n (%)	4 (3)	0 (0)	1 (2)	3 (6)
E/e'	17 <u>+</u> 6	16 <u>+</u> 5	17 <u>+</u> 7	18 ± 7
LVEDVi, mm/m ²	55 <u>+</u> 18	52 ± 13	51 ± 20	60 ± 18 $^+$
LVMI, g/m ²	168 ± 43	165 <u>+</u> 45	167 <u>+</u> 39	170 ± 45
LVEF, %	51 <u>+</u> 10	52 ± 8	52 ± 9	$48 \pm 11^{+}$
HFpEF, n (%)	90 (62)	19 (68)	46 (72)	25 (50) ⁺
HFmrEF, n (%)	26 (19)	6 (21)	10 (16)	10 (20)
HFrEF, n (%)	26 (19)	3 (11)	8 (12)	15 (30) ⁺
TAPSE, mm	18 <u>+</u> 5	18 ± 5	18 ± 4	17 <u>+</u> 5
sPAP, mmHg	40 <u>+</u> 9	40 ± 10	40 ± 9	41 <u>+</u> 9
Cardiorespiratory				
monitoring				
24-h AHI, events/h	12 (5–22)	3 (2–4)	12 (8–17) *	21 (11–31) * ^{,†}
Daytime AHI, events/h	7 (3–15)	1 (1–2)	8 (4–13) *	14 (7–22) * ^{,†}
Nighttime AHI, events/h	17 (9–31)	6 (4–9)	19 (13–29) *	29 (16–42) * ^{,†}
T90, min	20 (12–43)	11 (7–23)	20 (13–42)	28 (14–65) * ^{.†}

 Table 2
 Echocardiographic and cardiorespiratory monitoring parameters of the study population according to the breathing pattern

Values are mean \pm SD, median (inter-quartile range), or n (%).

AHI, apnoea–hypopnoea index; CAI, central apnoea index; LAVI, left atrial volume index diameter; LVEDVi, left ventricular end-diastolic volume index; LVEF, left ventricular ejection fraction; LVMI, left ventricular mass index; MR, mitral regurgitation; RVD, right ventricular diameter; sPAP, systolic pulmonary arterial pressure; TAPSE, tricuspid annular plane systolic excursion; T-90, percentage of time asleep with $SaO_2 < 90\%$.

*P < 0.05 vs. normal breathing; †P < 0.05 vs. obstructive approas.

may have influenced this result. Indeed, these patients, characterized by poorer outcomes at follow-up, had a substantially higher T-90 (almost double that of patients with ATTR-CA) despite a lower AHI.¹⁵ It is possible that mechanisms other than apnoeas or hypophoeas contributed to the desaturation burden in these patients, including more severe lung congestion (as suggested by higher NT-proBNP values and worse NYHA class), pulmonary hypertension, or other pulmonary comorbidities. Conversely, it can be hypothesized that different mechanisms contribute to the detrimental effects of breathing disorders in patients with ATTR-CA. Among these, the haemodynamic consequences of apnoeaand hyperventilation-related thoracic pressure swings may be particularly harmful in the case of restrictive cardiac physiology.²⁵ Additionally, sympathetic surges resulting from intermittent chemoreflex stimulation due to recurrent hypoxia and hypercapnia may contribute to adverse cardiac remodelling and increase the risk of life-threatening arrhythmias.^{26,27} Given these different findings, further studies investigating the pathophysiological consequences of breathing disorders in patients with ATTR-CA are necessary.

Similar to previous observations in HF,^{11,12,28} CAs were not only a sleep-related phenomenon in patients with ATTR-CA, but also a daytime occurrence, with at least moderate (AHI \geq 15 events/h) CA breathing disorders documented in 17% of the patients. Importantly, daytime AHI was associated with worse symptoms (higher NYHA class), greater neurohormonal activation, and ongoing cardiac damage, as indicated by higher plasma levels of NT-proBNP and hs-troponin T, as well as worse LV systolic and diastolic dysfunctions. These findings further underscored the importance of screening for breathing disorders not only during sleep time but also during the whole 24 h period in patients with HF secondary to different aetiologies, including ATTR-CA. However, contrary to previous reports in patients with HFrEF,^{11,12} daytime AHI was not an independent predictor of mortality in the current study. The low prevalence of severe forms (only 6% of patients had a daytime AHI \geq 30 events/h) might have underpowered the prognostic analysis for daytime AHI. Nevertheless, as chronically reduced cardiac output contributes to the sensitization of chemoreceptors, chemoreflex sensitivity may be lower in patients with ATTR-CA, whose LV systolic function is generally preserved. Considering that cardiac output is usually higher during the day,²⁹ this may contribute to the lower severity and prognostic value of daytime AHI. Conversely, the sleep-related reduction in cardiac output and increased filling pressure secondary to rostral fluid shift may exert opposing effects during the night in this subset, increasing circulatory delay and chemoreflex stimulation.²¹ This may explain the stronger association between nighttime AHI and mortality. Considering the possible therapeutic implications (as discussed in the following), further studies evaluating the determinants of daytime and nighttime CA in patients with ATTR-CA are needed.

While no studies have investigated the benefits of treating breathing disorders in ATTR-CA, different strategies have been tested in patients with HF.^{30,31} Although reducing the apnoea burden, mask-based ventilation systems failed to improve hard outcomes.^{32–34} Accordingly, in the latest ADVENT-HF trial, enrolling 731 patients with chronic HF (LVEF \leq 45%) and an AHI \geq 15 events/h, the adaptive servo-ventilation system did not affect the primary outcome (a composite of mortality, cardiovascular hospitalization, new onset atrial fibrillation or flutter, and appropriate cardioverter–defibrillator shock), irrespective of the prevailing apnoea type (CA or OA).³⁴ Alternative options are emerging

	24 h AHI		Daytime AHI		Nighttime AHI	
Variables	ho coefficient (95% Cl)	Р	ρ coefficient (95% Cl)	Р	ρ coefficient (95% Cl)	Р
Age	0.12 (-0.04-0.29)	0.127	0.14 (-0.02-0.29)	0.095	0.11 (-0.05-0.27)	0.184
BMI	0.07 (-0.12-0.25)	0.471	-0.02 (-0.20-0.17)	0.847	0.09 (-0.09-0.27)	0.346
NYHA class	0.19 (0.02-0.35)	0.025	0.22 (0.05-0.38)	0.008	0.12 (-0.05-0.29)	0.153
Atrial fibrillation	-0.09 (-0.25-0.09)	0.360	-0.06 (-0.23-0.19)	0.531	-0.10 (-0.28-0.07)	0.222
eGFR, mL/min/1.73 m ²	-0.18 (-0.36-0.01)	0.053	-0.20 (-0.37-(-0.01))	0.034	-0.16 (-0.34-0.02)	0.086
NT-proBNP, ng/L	0.30 (0.14–0.44)	<0.001	0.42 (0.26-0.56)	<0.001	0.17 (0.01–0.33)	0.042
NAC stage	0.18 (0.02-0.34)	0.029	0.25 (0.09-0.39)	0.003	0.11 (-0.06-0.27)	0.208
hs-troponin T, ng/L	0.25 (0.07-0.42)	0.008	0.28 (0.09-0.45)	0.003	0.19 (0.00-0.37)	0.047
Tafamidis	-0.09 (-0.25-0.08)	0.286	-0.11 (-0.27-0.05)	0.182	-0.04 (-0.21-0.12)	0.625
Beta-blockers	-0.10 (-0.07-0.27)	0.247	-0.08 (-0.08-0.26)	0.316	-0.08 (-0.09-0.25)	0.352
ACEi/ARB	-0.06 (-0.23-0.11)	0.452	-0.05 (-0.22-0.12)	0.585	-0.08 (-0.25-0.09)	0.368
MRA	0.01 (-0.16-0.18)	0.904	-0.02 (-0.19-0.15)	0.804	0.06 (-0.11-0.23)	0.501
SGLT2i	-0.05 (-0.22-0.12)	0.552	0.04 (-0.13-0.21)	0.657	-0.08 (-0.25-0.09)	0.334
LAVi, mL/m ²	0.15 (-0.02-0.33)	0.084	0.21 (0.04–0.37)	0.018	0.05 (-0.11-0.21)	0.539
MR severity	0.29 (0.13-0.44)	<0.001	0.35 (0.19–0.49)	<0.001	0.19 (0.02–0.35)	0.029
E/e'	0.14 (-0.03-0.33)	0.113	0.19 (0.02–0.36)	0.030	0.12 (-0.04-0.27)	0.139
LVMI	0.05 (-0.13-0.22)	0.601	0.12 (-0.05-0.29)	0.162	0.03 (-0.21-0.14)	0.706
LVEF, %	-0.22 (-0.38-(-0.06))	0.009	-0.18 (-0.34-(-0.01))	0.034	-0.16 (-0.32-0.01)	0.065
TAPSE, mm	-0.12 (-0.28-0.06)	0.172	-0.16 (-0.32-0.01)	0.056	0.03 (-0.15-0.20)	0.761
sPAP, mmHg	0.10 (-0.08-0.27)	0.267	0.09 (-0.08-0.27)	0.299	0.10 (-0.08-0.27)	0.267

 Table 3
 Clinical, humoral, and echocardiographic correlates of breathing disorders in patients with transthyretin cardiac amyloidosis

Abbreviations as in *Tables 1* and 2.



Figure 3 Severity of breathing disorders in survivors vs. non-survivors among patients with transthyretin cardiac amyloidosis. The severity of breathing disorders—expressed as the apnoea–hypopnoea index across the whole 24 h period, at daytime, or at nighttime—was higher in survivors vs. non-survivors.

to treat breathing disorders in a more rational and pathophysiologybased fashion. As regards OA, mandibular advancement devices seem promising to improve patient compliance,³⁵ while novel bioelectronic and pharmacological approaches are under investigation.³¹ As regards CA, while optimizing HF drug treatment should always represent the first step,³⁶ the pharmacological modulation of the overactive chemoreflex seems an intriguing opportunity, also considering its pathophysiological and independent prognostic role.^{26,27,37}

Study limitations

Considering the cross-sectional design of the study, the longitudinal changes in apnoea prevalence and severity over time remain to be investigated in patients with ATTR-CA, particularly in response to disease-modifying therapies (e.g. tafamidis) or tailored treatments (e.g. continuous positive airway pressure or oxygen). Within the study population, 16 patients with a nighttime AHI \geq 30 events/h and prevalent OA were referred for pulmonological evaluation to determine the



Figure 4 Prognostic value of severe breathing disorders in patients with transthyretin cardiac amyloidosis. In patients with transthyretin cardiac amyloidosis, a higher apnoea–hypopnoea index across the whole 24 h period, at daytime, and at nighttime was associated with a higher risk of all-cause death during follow-up.

appropriateness of specific therapeutic interventions. A domestic CPAP therapy was recommended to eight of these patients, with five initiating the treatment. However, the CPAP therapy was not a significant predictor of mortality at regression analysis. While this is likely due to the small number of patients treated, considering the findings of this study, the inclusion of patients with ATTR-CA should be considered in future studies on the treatment of breathing disorders. In the present study, unattended cardiorespiratory monitoring was used instead of in-hospital polysomnography. Nonetheless, although the use of this method did not allow for the assessment of arousal and sleep stages, it reduces costs and discomfort for the patients and provides data about the whole 24 h period, which could be of potential interest.^{11,12,16} Male sex, increased body mass index, and anatomical features of the neck and upper airways are the main determinants of OA in the general population,¹⁰ while increased chemoreflex gain, plant gain, and circulatory delay are the main determinants of CA in patients with HF.¹³ Although no substantial difference in sex distribution and body mass index was observed in our population among the groups, future dedicated studies are expected to investigate the pathophysiological determinants of breathing disorders in patients with ATTR-CA. In patients with HFmrEF and HFrEF, the prognostic impact of CA seems higher in female patients than in male patients.¹⁶ As in the present study, the majority of patients (91%) were male, and the existence of sex-related differences was not addressed and remains to be investigated in this subset. In this study, reflecting a contemporary real-life cohort, most of the patients had wild-type ATTR-CA; therefore, no comparison between wild-type and hereditary ATTR-CA was drawn. Considering that hereditary ATTR is a heterogeneous syndrome with a broad spectrum of clinical manifestations (cardiac, neurological, and mixed),³⁸ future studies should assess the prevalence and clinical significance of apnoeas in these patients. Because of the retrospective design of the study and the absence of a dedicated registry, the precise cause and time of death of the patients could not be retrieved. Considering the chronobiological features of breathing disorders and their link with life-threatening cardiovascular events, this analysis should be addressed by future prospective studies.

Conclusions

Breathing disorders including CA and OA are highly prevalent in patients with ATTR-CA, both at daytime and nighttime, likely contributing to disease severity and poor outcome. The findings of this study should therefore stimulate future research towards the following: (i) screen patients with ATTR-CA for breathing disorders to improve clinical severity characterization and risk prediction; (ii) evaluate the pathophysiological determinants of CA and OA in these patients and assess the effects of disease-modifying therapies on their prevalence, severity, and prognostic impact; and (iii) consider the inclusion of patients with ATTR-CA when designing future studies investigating the potential benefit of targeting breathing disorders through tailored interventions.

Supplementary material

Supplementary material is available at European Journal of Preventive Cardiology.

Author contribution

F.G., A.G., M.E., G.V., and C.P. contributed to the conception or design of the work. A.A., V.C., F.B., G.I., and E.D. contributed to the acquisition, analysis, or interpretation of data for the work. F.G., A.A., and V.C. drafted the article. A.G., M.E., G.V., and C.P. critically revised the article. All authors gave their final approval and agreed to be accountable for all aspects of work ensuring integrity and accuracy.

Funding

None declared.

Conflict of interest: No conflicts of interest to disclose.

Data availability

The data underlying this article are available upon reasonable request to the corresponding author.

References

- Kittleson MM, Maurer MS, Ambardekar AV, Bullock-Palmer RP, Chang PP, Eisen HJ, et al. Cardiac amyloidosis: evolving diagnosis and management: a scientific statement from the American Heart Association. *Circulation* 2020;**142**:e7–e22.
- Garcia-Pavia P, Rapezzi C, Adler Y, Arad M, Basso C, Brucato A, et al. Diagnosis and treatment of cardiac amyloidosis: a position statement of the ESC Working Group on Myocardial and Pericardial Diseases. *Eur Heart J* 2021;42:1554–1568.
- Rapezzi C, Aimo A, Barison A, Emdin M, Porcari A, Linhart A, et al. Restrictive cardiomyopathy: definition and diagnosis. Eur Heart J 2022;43:4679–4693.

- Bellavia D, Abraham TP, Pellikka PA, Al-Zahrani GB, Dispenzieri A, Oh JK, et al. Detection of left ventricular systolic dysfunction in cardiac amyloidosis with strain rate echocardiography. J Am Soc Echocardiogr 2007;20:1194–1202.
- AbouEzzeddine OF, Davies DR, Scott CG, Fayyaz AU, Askew JW, McKie PM, et al. Prevalence of transthyretin amyloid cardiomyopathy in heart failure with preserved ejection fraction. JAMA Cardiol 2021;6:1267–1274.
- Bhattacharya PT, Teruya SL, De Los Santos J, Helmke S, Maurer MS. Heart failure with reduced ejection fraction: an under-appreciated clinical phenotype of transthyretin cardiac amyloid (ATTR-CA). J Card Fail 2022;28:S35–S36.
- Emdin M, Aimo A, Rapezzi C, Fontana M, Perfetto F, Seferović PM, et al. Treatment of cardiac transthyretin amyloidosis: an update. Eur Heart J 2019;40:3699–3706.
- Aimo A, Vergaro G, Castiglione V, Fabiani I, Barison A, Gentile F, et al. Wild-type transthyretin cardiac amyloidosis is not rare in elderly subjects: the CATCH screening study. Eur J Prev Cardiol 2024;31:1410–1417.
- Cowie MR, Linz D, Redline S, Somers VK, Simonds AK. Sleep disordered breathing and cardiovascular disease: JACC state-of-the-art review. J Am Coll Cardiol 2021;78:608–624.
- Gottlieb DJ, Punjabi NM. Diagnosis and management of obstructive sleep apnea: a review. JAMA 2020;323:1389–1400.
- Emdin M, Mirizzi G, Giannoni A, Poletti R, Iudice G, Bramanti F, et al. Prognostic significance of central apneas throughout a 24-hour period in patients with heart failure. J Am Coll Cardiol 2017;70:1351–1364.
- Giannoni A, Gentile F, Sciarrone P, Borrelli C, Pasero G, Mirizzi G, et al. Upright Cheyne-Stokes respiration in patients with heart failure. J Am Coll Cardiol 2020;75:2934–2946.
- Giannoni A, Gentile F, Navari A, Borrelli C, Mirizzi G, Catapano G, et al. Contribution of the lung to the genesis of Cheyne-Stokes respiration in heart failure: plant gain beyond chemoreflex gain and circulation time. J Am Heart Assoc 2019;8:e012419.
- Borrelli C, Gentile F, Sciarrone P, Mirizzi G, Vergaro G, Ghionzoli N, et al. Central and obstructive apneas in heart failure with reduced, mid-range and preserved ejection fraction. Front Cardiovasc Med 2019;6:125.
- Bodez D, Guellich A, Kharoubi M, Covali-Noroc A, Tissot C-M, Guendouz S, et al. Prevalence, severity, and prognostic value of sleep apnea syndromes in cardiac amyloidosis. Sleep 2016;39:1333–1341.
- Gentile F, Borrelli C, Sciarrone P, Buoncristiani F, Spiesshoefer J, Bramanti F, et al. Central apneas are more detrimental in female than in male patients with heart failure. J Am Heart Assoc 2022;11:e024103.
- 17. Somers VK, White DP, Amin R, Abraham WT, Costa F, Culebras A, et al. Sleep apnea and cardiovascular disease: an American Heart Association/American College of Cardiology Foundation Scientific Statement from the American Heart Association Council for High Blood Pressure Research Professional Education Committee, Council on Clinical Cardiology, Stroke Council, and Council on Cardiovascular Nursing in collaboration with the National Heart, Lung, and Blood Institute National Center on Sleep Disorders Research (National Institutes of Health). J Am Coll Cardiol 2008;**52**:686–717.
- 18. Lang RM, Badano LP, Mor-Avi V, Afilalo J, Armstrong A, Ernande L, et al. Recommendations for cardiac chamber quantification by echocardiography in adults: an update from the American Society of Echocardiography and the European Association of Cardiovascular Imaging. J Am Soc Echocardiogr 2015;28:1–39.e14.
- Gillmore JD, Damy T, Fontana M, Hutchinson M, Lachmann HJ, Martinez-Naharro A, et al. A new staging system for cardiac transthyretin amyloidosis. Eur Heart J 2018;39: 2799–2806.
- Hosokawa K, Ando S-I, Tohyama T, Kiyokawa T, Tanaka Y, Otsubo H, et al. Estimation of nocturnal cardiac output by automated analysis of circulation time derived from polysomnography. Int J Cardiol 2015;181:14–16.

- Chenuel BJ, Smith CA, Skatrud JB, Henderson KS, Dempsey JA. Increased propensity for apnea in response to acute elevations in left atrial pressure during sleep in the dog. J Appl Physiol (1985) 2006;101:76–83.
- 22. Bitter T, Westerheide N, Prinz C, Hossain MS, Vogt J, Langer C, et al. Cheyne-Stokes respiration and obstructive sleep apnoea are independent risk factors for malignant ventricular arrhythmias requiring appropriate cardioverter-defibrillator therapies in patients with congestive heart failure. Eur Heart J 2011;**32**:61–74.
- Grimm W, Sosnovskaya A, Timmesfeld N, Hildebrandt O, Koehler U. Prognostic impact of central sleep apnea in patients with heart failure. J Card Fail 2015;21:126–133.
- Oldenburg O, Wellmann B, Buchholz A, Bitter T, Fox H, Thiem U, et al. Nocturnal hypoxaemia is associated with increased mortality in stable heart failure patients. Eur Heart J 2016;37:1695–1703.
- Giannoni A, Raglianti V, Taddei C, Borrelli C, Chubuchny V, Vergaro G, et al. Cheyne-stokes respiration related oscillations in cardiopulmonary hemodynamics in patients with heart failure. Int J Cardiol 2019;289:76–82.
- Giannoni A, Gentile F, Buoncristiani F, Borrelli C, Sciarrone P, Spiesshoefer J, et al. Chemoreflex and baroreflex sensitivity hold a strong prognostic value in chronic heart failure. JACC Heart Fail 2022;10:662–676.
- Giannoni A, Borrelli C, Gentile F, Sciarrone P, Spießhöfer J, Piepoli M, et al. Autonomic and respiratory consequences of altered chemoreflex function: clinical and therapeutic implications in cardiovascular diseases. Eur J Heart Fail 2023;25:642–656.
- Brack T, Thüer I, Clarenbach CF, Senn O, Noll G, Russi EW, et al. Daytime Cheyne-Stokes respiration in ambulatory patients with severe congestive heart failure is associated with increased mortality. *Chest* 2007;**132**:1463–1471.
- Veerman DP, Imholz BP, Wieling W, Wesseling KH, van Montfrans GA. Circadian profile of systemic hemodynamics. *Hypertension* 1995;26:55–59.
- Gentile F, Sciarrone P, Buoncristiani F, Castiglione V, Bramanti F, Iudice G, et al. [Central apneas and cardiovascular diseases]. G Ital Cardiol (Rome) 2023;24:701–710.
- Castiglione V, Morfino P, Gentile F, Airò E, Passino C, Giannoni A, et al. [Obstructive sleep apneas and cardiovascular diseases]. G Ital Cardiol (Rome) 2023;24:979–989.
- Bradley TD, Logan AG, Kimoff RJ, Sériès F, Morrison D, Ferguson K, et al. Continuous positive airway pressure for central sleep apnea and heart failure. N Engl J Med 2005; 353:2025–2033.
- Cowie MR, Woehrle H, Wegscheider K, Angermann C, d'Ortho M-P, Erdmann E, et al. Adaptive servo-ventilation for central sleep apnea in systolic heart failure. N Engl J Med 2015;373:1095–1105.
- 34. Bradley TD, Logan AG, Filho GL, Kimoff RJ, Cantolla JD, Arzt M, et al. Adaptive servoventilation for sleep-disordered breathing in patients with heart failure with reduced ejection fraction (ADVENT-HF): a multicentre, multinational, parallel-group, openlabel, phase 3 randomised controlled trial. *Lancet Respir Med* 2024;**12**:153–166.
- Emdin M, Gentile F, Giannoni A. Mandibular advancement versus CPAP for obstructive sleep apnea: open your mouth wide and breathe. J Am Coll Cardiol 2024;83:1773–1774.
- 36. Passino C, Sciarrone P, Vergaro G, Borrelli C, Spiesshoefer J, Gentile F, et al. Sacubitril-valsartan treatment is associated with decrease in central apneas in patients with heart failure with reduced ejection fraction. Int J Cardiol 2021;330:112–119.
- Giannoni A, Borrelli C, Mirizzi G, Richerson GB, Emdin M, Passino C. Benefit of buspirone on chemoreflex and central apnoeas in heart failure: a randomized controlled crossover trial. *Eur J Heart Fail* 2021;23:312–320.
- Caponetti AG, Sguazzotti M, Accietto A, Saturi G, Ponziani A, Giovannetti A, et al. Characterization and natural history of different phenotypes in hereditary transthyretin amyloidosis: 40-year experience at a single Italian referral centre. Eur J Prev Cardiol 2024; 31:866–876.